Author’s response to reviews

Title: Thymoma complicated with myasthenia gravis and Good syndrome – a therapeutic conundrum: A case report

Authors:

Shiran Paranavitane (shiran_p@live.com)
Sumana Handagala (handagala@yahoo.com)
Rajiva De Silva (nilhandesilva@gmail.com)
Thashi Chang (thashichang@gmail.com)

Version: 4 Date: 08 Sep 2019

Author’s response to reviews:

1) The follow-up period has been mentioned in the manuscript (6 months). It is highlighted in red.

2) The reviewers have requested the radiological investigations in addition to their description. However, these HAVE NOT BEEN INCLUDED. The reason for this being that the purpose of this case report is to highlight an important clinical conundrum which was encountered and provide guidance on managing such a patient to other clinicians. Therapeutics and logical clinical reasoning are the highlights of this manuscript and not radiological investigations.